Author’s response to reviews

Title: Carotid-cavernous fistula (CCF) presenting as paroxysmal painful ophthalmoplegia

Authors:
Shan Li (150406830@qq.com)
Bin Feng (38020104@qq.com)
Yabo Feng (531503307@qq.com)
Zaiying Pang (doctor_pang@163.com)
Youting Lin (lincyuting@hotmail.com)

Version: 1 Date: 08 Jan 2019

Author’s response to reviews:

Dear Editor:

Thank you very much for your letter and the comments about our paper submitted to BMC Ophthalmology. The manuscript entitled “Carotid-cavernous fistula (CCF) presenting as paroxysmal painful ophthalmoplegia” have been revised according to the reviewers’ comments, and we wish it to be reconsidered for publication in BMC Ophthalmology.

A list of changes and responses to reviewers are as follows:

Reviewer reports:

George Kalantzis, M.D., FRCOphtht (Reviewer 1): This is an interesting case of CCF. However, the text is unnecessary long and requires extensive editing in English language.

Lepša D. Žoric (Reviewer 2): Good manuscript, proper history and way of thinking.

To Reviewer 1:

Thank you very much for pointing out the problems in our manuscript. We have revised it according to your recommendations. The text has been condensed, and edited by native English-speaking editors at Nature Research Editing Service. The certification has been uploaded.
We hope that all these changes fulfill the requirements to make the manuscript acceptable for publication in BMC Ophthalmology.

Looking forward to hearing from you soon.

Sincerely yours,

Youting Lin